Author's response to reviews

Title: Reducing Depression in Older Home Care Clients: Design of a Prospective Study of a Nurse-Led Interprofessional Mental Health Promotion Intervention

Authors:

Maureen F Markle-Reid (mreid@mcmaster.ca)
Carrie McAiney (mcaineyc@mcmaster.ca)
Dorothy Forbes (dorothy.forbes@ualberta.ca)
Lehana Thabane (thabanl@mcmaster.ca)
Maggie Gibson (Maggie.Gibson@sjhc.london.on.ca)
Jeffrey S Hoch (jeffrey.hoch@utoronto.ca)
Gina Browne (browneg@mcmaster.ca)
Thomas Peirce (tom.peirce@hnhb.ccac-ont.ca)
Barbara Busing (barbara.busing@hnhb.ccac-ont.ca)

Version: 6 Date: 24 August 2011

Author's response to reviews:

August 24, 2011
Rachel Neilan, MSc
Executive Editor
BMC Geriatrics

Dear Ms. Neilan:

RE: MS: 4046258775487766
Reducing Depression in Older Home Care Clients: Design of a Prospective Study of a Nurse-Led Interprofessional Mental Health Promotion Intervention

Thank you for sending the revised version of our manuscript. We have checked the manuscript again and have made a few minor formatting changes to the paper. Thank you again for the opportunity to publish in BMC Geriatrics.

Kind Regards,

Maureen Markle-Reid, RN, MScN, PhD